Author's response to reviews

Title: Excision of sympathetic ganglia and the rami communicantes with histologic confirmation offers better early and late outcomes in Video assisted thoracoscopic sympathectomy.

Authors:

Sridhar Rathinam (srathinam@rcsed.ac.uk)
Prakash Nanjaiah (drprks@yahoo.com)
Sivakumar Sivalingam (sivkuma71@yahoo.com)
Pala B Rajesh (pala.rajesh@heartofengland.nhs.uk)

Version: 4 Date: 12 August 2008

Author's response to reviews:

Dear Sir,
We are delighted to hear that our work has been accepted for publication and have herewith attached the revised manuscript.
The manuscript has been formatted with double spacing.
A figures legend has been added
A uniform style has been used for Video Assisted Thoracoscopic Sympathectomy
The Corresponding authors phone number has been changed.
Figure 1 has been referenced in the text.
I hope that these changes meet with your approval
Kind Regards
S Rathinam